\begin{table}%t1
\caption{\label{t:fits}Observations: inner disks in the low/hard state.}
%\centerline
{\begin{tabular}{@{\extracolsep{\fill}}llllcl}
\hline\hline
&Source & Outburst phase & Thermal component & Inner disk? & References\\
&  & &  $L_{\rm{disk}}/L_{\rm{Edd}}$ & & \\
\hline\noalign{\smallskip}
& GX 339-4 & outburst 2004, rise & $5\times10^{-2}$ &  present& 1, 2\\
&GX 339-4 & outburst 2007, decline & $2.3\times10^{-2} $ &  present &3\\
&GX 339-4 & outburst 2007, decline & $8\times10^{-3} $
  & present (weaker evidence) & 3\\
&J1753.5-0127 & outburst 2006, decline & $3\times10^{-3} $ &
present &4\\
&XTE J1817-330 & outburst 2006, decline & $1.2\times10^{-3} $
& present &5\\
&XTE J1118+480 &  hard state outburst 2000 &$10^{-3} $   & present & 6, 7\\
&GRO J1655-40 & outburst 2005, rise  &  $10^{-3}$  & small
significant evidence &5\\
&V4641 SGR & outburst 1999, rise & $1.5\times10^{-3}$ ?&present &8\\
&XTE J1650-500 &outburst 2002, rise ? & $10^{-3}$ &present & 9, 10\\
& & outburst 2002, late decline &$ \le$$10^{-4}$ & no evidence &11\\
&4U 1543-47 & outburst 2002, late decline  & 
  $10^{-5}{-}10^{-6}$  &  some evidence &12, 13\\
&Cygnus X-1  & hard intermediate state 2001&$7\times10^{-2}$ & present&
10, 14\\
&SAX J1711.6-3808 &outburst 2001, decline & .~~.~~. &present  ? & 15\\
&XTE J1908+094 &outburst 2002 & .~~.~~.&present ? & 16\\
&V404 Cyg   & quiescence & $\le$$10^{-6}$ &no evidence &17\\
&XTE 1550 &&&&\\
&and H1743-322 & quiescence & $\le$$10^{-6}$ & no evidence& 18\\
\hline
\end{tabular}}
\medskip
References: 1) Miller et~al. (\cite{Miller2006b}b), 2) Reis et~al. (\cite{Reis2008}), 3) Tomsick et~al. (\cite{Tomsick2008}), 4) Miller et~al. (\cite{Miller2006a}a), 5) Rykoff et~al. (\cite{Rykoff2007}), 6)
McClintock et~al. (\cite{McClintock2001}), 7)  Reis et~al. (\cite{Reis2009}), 8) Miller et~al. (\cite{Miller2002c}c), 9) Miller et~al. (\cite{Miller2002a}a), 10) Miller et~al. (\cite{Miller2009}), 11) Tomsick et~al. (\cite{Tomsick2004}),  12) Kalemci et~al. (\cite{Kalemci2005}), 13) La Palombara \& Mereghetti (\cite{LaPalombara2005}), 14) Miller et~al. (\cite{Miller2002b}b), 15) in 't Zand et~al. (\cite{Zand2002a}a), 16) in 't Zand et~al. (\cite{Zand2002b}b), 17)  Bradley et~al. (\cite{Bradley2007}), 18) Corbel et~al. (\cite{Corbel2006}). 
For a discussion of details and sources with peculiar dim soft
  states, GRS 1758-258 and 1E 1740.7-2942, see text.
\end{table}